Author’s response to reviews

Title: Calibration of dentists for Caries Management by Risk Assessment Research in a Practice Based Research Network - CAMBRA PBRN

Authors:

Peter Rechmann (Peter.Rechmann@ucsf.edu)
Bonnie Jue (bonniejue@sbcglobal.net)
William Santo (William.Santo@ucsf.edu)
Beate Rechmann (Beate.Rechmann@ucsf.edu)
John Featherstone (John.Featherstone@ucsf.edu)

Version: 2 Date: 20 Oct 2017

Author’s response to reviews:

Ref.: Manuscript OHEA-D-17-00062R1

Calibration of dentists for Caries Management by Risk Assessment Research in a Practice Based Research Network - CAMBRA PBRN.

Dear Editor,

Dear Reviewers,

Dear Drs. Sonia Makhija, and Fernanda Ferreira,

Thank you for reviewing our revised manuscript, “Calibration of Dentists for Caries Management by Risk Assessment Research in a Practice Based Research Network - CAMBRA PBRN”!

Thanks for accepting the latest version for publication, if editorial requirements are fulfilled.
Here are our detailed responses:

Reply to reviewer 1 – Dr. Sonia Makhija
- Reviewer Recommendation: Accept without revision

Thank you!

Reply to reviewer 2 - Dr. Fernanda Ferreira
- Reviewer Recommendation: Accept without revision

I congratulate the authors for doing a good job in answering satisfactory all the topics addressed.

Thanks you so much for your kind comment!

Editor Comments:

1. Please remove the response to reviewers from the file inventory, as it is no longer needed at this stage of the editorial process.

We removed the response to reviewers.

2. Please list the full names, institutional addresses and email addresses for all authors on the title page.

All full names, and addresses are listed for each author
3. In the Funding section, please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

We added to the Funding section: “The funding body had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.”

4. Please upload a clean version of the manuscript without tracked changes or highlighting and then remove any highlighted versions from the inventory, as these are no longer required at this stage of the editorial process.

A clean version of the manuscript has been uploaded.

Sincerely yours,

The authors